Author's response to reviews

Title: Peripheral capillary nonperfusion in asymptomatic Waldenstrom's macroglobulinemia

Authors:

Chryssanthi Koutsandrea (ancode@otenet.gr)
Athanassios Kotsolis (akotso@hotmail.com)
Ilias Georgalas (igeorgalas@yahoo.com)
Dimitris Papaconstantinou (dpapaconstantinou@hotmail.com)
Ioannis Ladas (idladas@gmail.com)

Version: 3 Date: 3 November 2010

Author's response to reviews:

Re: “MS: 1535410802394906
Peripheral capillary nonperfusion in asymptomatic Waldenstrom's macroglobulinemia”

Dear Editor,

Thank you for your interest in our manuscript and your comments which really add to its clarity.

Please find attached the revised submission of the manuscript entitled “Peripheral capillary nonperfusion in asymptomatic Waldenstrom's macroglobulinemia” according with all the instructions of the reviewers.

The detailed response to the comments follows:

Reviewer’s report:
The first sentence in the conclusion should be changed into: "The most frequent reported retinal findings in hyperviscosity syndrome...." Otherwise it contradicts the next sentence.
Response: The first level of the conclusion has now been changed.

Sincerely yours,
Ilias Georgalas, MD
Consultant Ophthalmic Surgeon